Author’s response to reviews

Title: Comparison of Visual Results and Higher-order Aberrations After Small Incision Lenticule Extraction (SMILE): High Myopia vs. Mild to Moderate Myopia

Authors:

hongying jin (hongyingj@163.com)

Ting Wan (64230519@qq.com)

Fang Wu (fangwuu@126.com)

Ke Yao (xlren@zju.edu.cn)

Version: 3 Date: 20 Jun 2017

Author’s response to reviews:

Dear editor,

We have modified the paper according to the review’s suggestions. We use red fonts in the paper to highlight the modification.

1. Thank you for providing a trial registration number. Please add the number to the end of the abstract section within the main manuscript. Please also include the date of registry along with the phrase ‘retrospectively registered’.

Answer: We add the number to the end of the abstract section. We include the data of registry along with the phrase ‘retrospectively registered’.

2. Thank you for providing an “Authors’ contributions” subheading for the Declarations section. However, please provide more justification for the contributions of authors. Currently, the contributions of Ke Yao do not automatically qualify for authorship. Please amend your manuscript accordingly.
Authors’ contributions

HYJ conceived of the study and drafted the manuscript. TW collected the data and revised the manuscript. FW collected the data and involved in drafting the manuscript. HYJ performed the statistical analysis. KY participated in the design of the study and critically revised the manuscript. All authors read and approved the final manuscript.

Yours Sincerely

Yao Ke, MD
Professor and Chief
Eye Institute of Zhejiang University
Eye Center, Second Affiliated Hospital, College of Medicine, Zhejiang University
Address: No. 88 Jiefang Road, Hangzhou 310009, China
E-mail address: xlren@zju.edu.cn.